

Citation: Hayes R, Floyd S, Schaap A, Shanaube K, Bock P, Sabapathy K, et al. (2017) A universal testing and treatment intervention to improve HIV control: One-year results from intervention communities in Zambia in the HPTN 071 (PopART) cluster-randomised trial. PLoS Med 14(5): e1002292. https://doi.org/10.1371/journal. pmed.1002292

Academic Editor: Nicola Low, University of Bern, SWITZERLAND

Received: November 17, 2016

Accepted: March 27, 2017

Published: May 2, 2017

Copyright: © 2017 Hayes et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: We have provided in the <u>S1 Data</u> and <u>S2 Data</u> supporting information files aggregate data that can be used to replicate the main analyses presented in this paper. These aggregate data comprise counts of relevant indicators stratified by community, gender and age-group.

Funding: HPTN 071 is sponsored by the National Institute of Allergy and Infectious Diseases (NIAID)

RESEARCH ARTICLE

A universal testing and treatment intervention to improve HIV control: One-year results from intervention communities in Zambia in the HPTN 071 (PopART) cluster-randomised trial

Richard Hayes¹*, Sian Floyd¹, Ab Schaap^{1,2}, Kwame Shanaube², Peter Bock³, Kalpana Sabapathy¹, Sam Griffith⁴, Deborah Donnell⁵, Estelle Piwowar-Manning⁶, Wafaa El-Sadr⁷, Nulda Beyers³, Helen Ayles^{2,8}, Sarah Fidler⁹, for the HPTN 071 (PopART) Study Team¹

 Department of Infectious Disease Epidemiology, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Zambart, University of Zambia School of Medicine, Lusaka, Zambia, 3 Desmond Tutu TB Centre, Department of Paediatrics and Child Health, University of Stellenbosch, Stellenbosch, South Africa, 4 FHI 360, HIV Prevention Trials Network, Durham, North Carolina, United States of America,
HIV Prevention Trials Network Statistical and Data Management Center, Statistical Center for HIV/AIDS Research and Prevention, Seattle, Washington, United States of America, 6 HIV Prevention Trials Network Laboratory Center, Johns Hopkins University School of Medicine, Baltimore, Maryland, United States of America, 7 Mailman School of Public Health, Columbia University, New York, New York, United States of America, 8 Department of Clinical Research, London School of Hygiene & Tropical Medicine, London, United Kingdom, 9 HIV Clinical Trials Unit, Imperial College London, London, United Kingdom

¶ Membership of the HPTN 071 (PopART) Study Team is provided in the Acknowledgments. * richard.hayes@lshtm.ac.uk

Abstract

Background

The Joint United Nations Programme on HIV/AIDS (UNAIDS) 90-90-90 targets require that, by 2020, 90% of those living with HIV know their status, 90% of known HIV-positive individuals receive sustained antiretroviral therapy (ART), and 90% of individuals on ART have durable viral suppression. The HPTN 071 (PopART) trial is measuring the impact of a universal testing and treatment intervention on population-level HIV incidence in 21 urban communities in Zambia and South Africa. We report observational data from four communities in Zambia to assess progress towards the UNAIDS targets after 1 y of the PopART intervention.

Methods and findings

The PopART intervention comprises annual rounds of home-based HIV testing delivered by community HIV-care providers (CHiPs) who also support linkage to care, ART retention, and other services. Data from four communities in Zambia receiving the full intervention (including immediate ART for all individuals with HIV) were used to determine proportions of participants who knew their HIV status after the CHiP visit; proportions linking to care and initiating ART following referral; and overall proportions of HIV-infected individuals who knew their status (first 90 target) and the proportion of these on ART (second 90 target),

under Cooperative Agreements UM1-AI068619, UM1-AI068617, and UM1-AI068613, with funding from the U.S. President's Emergency Plan for AIDS Relief (PEPFAR). Additional funding is provided by the International Initiative for Impact Evaluation (3ie) with support from the Bill & Melinda Gates Foundation, as well as by NIAID, the National Institute on Drug Abuse (NIDA) and the National Institute of Mental Health (NIMH), all part of the U. S. National Institutes of Health (NIH). The funders had no role in data collection, data analysis, or the decision to publish this manuscript. 3ie reviewed and provided non-binding comments on the draft manuscript prior to journal submission. The trial sponsor, NIH, reviewed and approved the original design of the HPTN 071 study and approves all changes to the study protocol.

PLOS MEDICINE

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: ANRS 12249 TasP trial, Agence Nationale de Recherche sur le Sida et les Hépatites Virales 12249 Treatment as Prevention trial; ART, antiretroviral therapy; CHiP, community HIV-care provider; EDC, electronic data capture; TB, tuberculosis; UNAIDS, Joint United Nations Programme on HIV/AIDS; UTT, universal testing and treatment. pre- and post-intervention. We are not able to assess progress towards the third 90 target at this stage of the study. Overall, 121,130 adults (59,283 men and 61,847 women) were enumerated in 46,714 households during the first annual round (December 2013 to June 2015). Of the 45,399 (77%) men and 55,703 (90%) women consenting to the intervention, 80% of men and 85% of women knew their HIV status after the CHiP visit. Of 6,197 HIV-positive adults referred by CHiPs, 42% (95% CI: 40%-43%) initiated ART within 6 mo and 53% (95% CI: 52%–55%) within 12 mo. In the entire population, the estimated proportion of HIVpositive adults who knew their status increased from 52% to 78% for men and from 56% to 87% for women. The estimated proportion of known HIV-positive individuals on ART increased overall from 54% after the CHiP visit to 74% by the end of the round for men and from 53% to 73% for women. The estimated overall proportion of HIV-positive adults on ART, irrespective of whether they knew their status, increased from 44% to 61%, compared with the 81% target (the product of the first two 90 targets). Coverage was lower among young men and women than in older age groups. The main limitation of the study was the need for assumptions concerning knowledge of HIV status and ART coverage among adults not consenting to the intervention or HIV testing, although our conclusions were robust in sensitivity analyses.

Conclusions

In this analysis, acceptance of HIV testing among those consenting to the intervention was high, although linkage to care and ART initiation took longer than expected. Knowledge of HIV-positive status increased steeply after 1 y, almost attaining the first 90 target in women and approaching it in men. The second 90 target was more challenging, with approximately three-quarters of known HIV-positive individuals on ART by the end of the annual round. Achieving higher test uptake in men and more rapid linkage to care will be key objectives during the second annual round of the intervention.

Trial registration

ClinicalTrials.gov NCT01900977

Author summary

Why was this study done?

- Epidemiological and modelling studies have shown that universal testing and treatment (UTT) for HIV, whereby everyone in the community should know their HIV status and all HIV-infected individuals should be offered immediate antiretroviral therapy (ART), may steeply reduce the incidence of new HIV infections in generalised epidemics in sub-Saharan Africa.
- However, it is not known if such UTT interventions can be implemented on a large scale in resource-poor settings, or what coverage can be achieved in practice.

What did the researchers do and find?

- We are carrying out a large community-randomised trial, the HPTN 071 (PopART) trial, to measure the impact of a UTT intervention on HIV incidence at the population level over three years in 21 urban communities in Zambia and South Africa.
- We report here on the uptake of the intervention achieved in four Zambian communities after the first annual round of the intervention, which is delivered house-to-house by trained lay health workers.
- Uptake of HIV testing with the lay health workers was high, and we found that after one year, an estimated 78% of men and 87% women who were HIV-positive knew their HIV status. Among known HIV-positive adults, around 73% were on ART by the end of the annual round.
- The estimated overall proportion of HIV-positive adults on ART, irrespective of whether they knew their status, increased from 44% to 61% following the intervention.
- The main challenges were that many men could not be contacted during home visits, that linkage to care and initiation of ART were slower than anticipated, and that overall coverage was lower in young people.

What do these findings mean?

- The Joint United Nations Programme on HIV/AIDS (UNAIDS) 90-90-90 targets require that, by 2020, 90% of HIV-positive individuals know their status (first 90 target), 90% of known HIV-positive individuals are on ART (second 90 target), and 90% of those have durable viral suppression (third 90 target).
- We have shown that after the first annual round of a community-wide household-based intervention, knowledge of HIV-positive status was close to the first 90 target in women, and approaching it in men. There was a substantial increase in the proportion of known HIV-positive individuals on ART, although the second 90 target has not yet been reached.
- The repeated annual rounds of household visits are a strength of the PopART intervention, and, based on the findings of the current analysis, future rounds will focus on key challenges, including reaching men and young people and achieving more rapid linkage to care and ART initiation.

Introduction

While there have been some reductions globally in the incidence of new HIV infections, incidence and prevalence remain at unacceptably high levels in many parts of sub-Saharan Africa. This is especially the case in southern Africa, where many regions are still experiencing severe generalised epidemics. Thanks to the large-scale rollout of antiretroviral therapy (ART), mortality from HIV has fallen [1]. HIV incidence has not fallen to the same extent, however, and the number of new HIV infections each year exceeds the number of deaths, meaning that the total number of HIV-infected individuals continues to increase [1]. Unless effective ways can be found of steeply reducing incidence, it will become increasingly difficult to sustain ART services for all who need them.

Wide-scale provision of ART is now recognised as a key preventive intervention for HIV control. The concept of treatment as prevention is based on evidence showing that HIV transmission is strongly correlated with viral load, and that the risk of transmission is very low if virus is undetectable [2–5]. The HPTN 052 trial showed definitively that early initiation of ART reduced HIV transmission by an estimated 96% [3]. Mathematical modelling has shown that if a large enough proportion of the HIV-positive population is diagnosed and started on treatment, with good enough retention and adherence to achieve high rates of viral suppression, this could lead to steep reductions in HIV incidence and potentially to the long-term elimination of HIV as a public health problem [6–9]. Observational data from a range of settings have demonstrated correlations between ART coverage at the population level and reductions in HIV incidence [10–14].

Based on this evidence, the Joint United Nations Programme on HIV/AIDS (UNAIDS) has promulgated a set of targets aimed at ensuring that a high proportion of the HIV-infected population is diagnosed, on ART, and virally suppressed, with the aim of reducing HIV incidence as well as protecting the health of HIV-infected individuals [15,16]. The 90-90-90 targets are to be achieved by 2020 and require that 90% of all people living with HIV know their status, that 90% of all people diagnosed with HIV receive sustained ART, and that 90% of all people receiving ART have durable viral suppression [15]. Taken together, meeting these targets would imply that approximately 73% of HIV-infected individuals would be virally suppressed. An even more ambitious set of 95-95-95 targets is to be achieved by 2030, with the aim of substantially reducing HIV incidence and prevalence by then [16].

While it may be useful to have aspirational targets for service coverage, it is not known whether such targets can be achieved on a large scale at the population level, or what are the most effective and cost-effective methods of achieving them. Changes to WHO guidelines to recommend immediate initiation of ART for all people living with HIV, irrespective of CD4 count, will remove one barrier to achieving the second 90 target. However, there are many other obstacles and challenges to ensuring universal testing and treatment (UTT), including the need to scale up HIV testing in the community so that everyone knows their current HIV status, and the need to ensure that all HIV-infected individuals are linked to care and commence ART rapidly, with high rates of retention, adherence, and viral suppression.

The HPTN 071 (PopART) trial is implementing a combination prevention programme in Zambia and South Africa aiming to achieve UTT through a house-to-house service delivered by a cadre of community health workers, and to measure its effectiveness and cost-effectiveness in reducing HIV incidence at the population level [17]. In this paper, we examine findings from the trial communities receiving the PopART UTT intervention in Zambia after the first year of the intervention. We examine what coverage has been achieved following the first annual round, what difference this has made to key indicators of uptake, and how close we have come to the first two of the 90-90-90 targets, which are important process outcomes for the trial. We also ask what are the most important barriers to meeting these targets. The impact of the PopART intervention on the primary outcome (HIV incidence) will be reported following trial completion in 2018.

Methods

Trial design

Full details of the HPTN 071 (PopART) trial and intervention have been presented previously [17]. In brief (Fig 1), 21 urban and peri-urban communities in Zambia and South Africa (each

3 arm cluster-randomised trial with 21 communities



Fig 1. Overview of design of the HPTN 071 (PopART) trial. ART, antiretroviral therapy; TB, tuberculosis.

https://doi.org/10.1371/journal.pmed.1002292.g001

community defined as the catchment population of a government health facility at which services for HIV care and ART provision were already in place) were formed into seven matched triplets (four in Zambia and three in South Africa) and randomly allocated to three study arms. Arm A is receiving the full PopART intervention including immediate ART irrespective of CD4 count. Arm B is receiving the full intervention except that ART is provided according to current national treatment guidelines. Arm C is a control arm, which continues to receive existing standard of care including ART according to current national treatment guidelines.

To measure the impact of the intervention, a population cohort comprising a random sample of adults aged 18-44 y (approximately 2,000 per community) has been enrolled. Following a baseline survey, the cohort will be followed up at annual intervals for 3 y. The primary objective of the trial is to measure the population-level impact of the intervention by comparing HIV incidence over 36 mo between the three study arms.

Here we report data on the uptake and coverage of the UTT intervention after the first annual round of intervention in the entire population of our Arm A communities in Zambia. During the first annual round, some challenges were encountered in the collection of intervention data from our South African communities. In particular, there was some under-recording of community members who were absent or declined to participate in the intervention or who declined an HIV test. These problems have since been resolved, and we plan to report uptake and coverage in South Africa during later rounds of intervention in due course.

Intervention

The PopART intervention is a combination prevention package with several components. A cadre of community HIV-care providers (CHiPs) provides house-to-house services for all members of the community, including annual HIV testing and counselling; linkage to care at the local government health facility and support for adherence and retention in care for those identified as HIV-positive; promotion of voluntary medical male circumcision (VMMC) for HIV-negative uncircumcised men and referral to local VMMC services; promotion of prevention of mother-to-child transmission (PMTCT) services for pregnant women; referral of those reporting symptoms of sexually transmitted infections for diagnosis and treatment at the health facility; screening for symptoms of tuberculosis (TB) and linkage to care for those diagnosed with TB; and promotion and provision of condoms. ART is provided as a routine service by the health facility.

CHiPs work in pairs, with each pair serving a zone consisting of around 500 households. They visit all households in their zone to provide annual testing and re-testing using HIV rapid tests, as well as making additional visits as necessary to complete intervention provision and follow-up. They record basic data on the household and all household members on an electronic data capture (EDC) device, with more detailed data (e.g., on HIV test history, ART status, and uptake of HIV testing and other services provided by CHiPs) recorded for adults aged 18 y and above who consent to receive the intervention. The EDC populates a CHiP database, which is used to obtain process measures on the uptake of the intervention. There is no fixed schedule for follow-up visits during the annual round, but special efforts are made to follow up all participants known to be HIV-positive to support them to link to and stay in HIV care, as well as to record information on these outcomes.

The PopART intervention will be delivered in four annual rounds. In this paper, we report the outcome of the first annual round (delivered from December 2013 to June 2015, with some variation as to when services were rolled out in communities) in the four Arm A communities in Zambia, to determine how close a UTT intervention can come to meeting the first two of the 90-90-90 targets after a single year of intervention. While services including testing and linkage to care are offered for all household members, ethical approval for data collection during the first annual round was obtained only for adults aged 18 y and over, and so analyses are restricted to this age group.

Statistical methods

The statistical analysis plan for the parent HPTN 071 (PopART) study is currently under development and review, and had not been finalised at the time that the process data analysis reported in this paper was carried out. The indicators of intervention coverage used for the current analyses were determined at the start of the trial, but when the UNAIDS 90-90-90 targets were announced at a later stage (September 2014), our analyses were extended to provide estimates of coverage against these targets.

Data from the CHiP database were first used to populate a cascade of care diagram showing the proportions completing each step required if individuals are to know their HIV status and initiate ART. Separate estimates were obtained for men and women, and the data were pooled across all four communities in Arm A. First, we calculated the proportion of households visited during the first annual round, with the denominator (number of households) obtained from a census carried out prior to the trial, and, second, we calculated the proportion of visited households that consented both to the intervention being explained and to enumeration (listing) of all household members on the EDC device. Third, we calculated the proportion of enumerated adult members who were contacted and consented to take part in the PopART intervention and, of these, the proportion who knew their HIV status following the CHiP visit because they

accepted the offer of an HIV test from the CHiP, self-reported as HIV-positive, or self-reported as HIV-negative based on a test within the previous 3 mo. Fourth, we computed the number and proportion of "known HIV-positive" adults who were referred to care by the CHiP, overall and with restriction to those who had never previously registered for HIV care. Finally, we computed the proportion of those referred who had linked to care and commenced ART by 6 mo and 12 mo after referral by a CHiP. Because individuals were referred at different times, with differing durations of follow-up, these proportions were estimated by Kaplan–Meier "time to event" analysis, using retrospective information from follow-up visits by CHiPs up to mid-2016 but with follow-up right-censored at the end of the first annual round (30 June 2015). The corresponding survival plots were obtained from the same analysis.

This cascade of care focuses on the success of the intervention in contacting adults, achieving high rates of participation among those contacted, achieving high rates of test uptake among those who consent to participate, and linking to care those HIV-positive adults who are not already on ART. We carried out further analyses to obtain a more complete picture of the overall proportion of adults who knew their HIV status, the overall proportion known to be HIV-positive, and the proportion of known HIV-positive adults who were in HIV care or on ART prior to the intervention, and to determine how much these proportions were increased by the intervention. Using the CHiP database, among those consenting to the PopART intervention, we computed the proportion who knew their HIV status (and the proportion known to be HIV-positive) among all adults, before and after the CHiP visit. Taking those who were known HIV-positive after the CHiP visit, as well as at the end of the first annual round of intervention (using as the denominator those still resident in the same zone of the community according to the most recent follow-up information, based on follow-up data collected on or before 30 June 2015).

Finally, we used these data to estimate coverage against the first two of the 90-90-90 targets: proportion of individuals with knowledge of HIV-positive status and proportion of known HIV-positive individuals on ART. All calculations were stratified by sex, community, and age group. Full details of the computations of these estimates are given in <u>S1-S4</u> Tables, illustrated using data for women in one age group. We present coverage estimates separately for men and women of all ages, and also by age group.

We first estimated coverage among all adults who consented to the PopART intervention. To estimate the proportion of HIV-positive individuals who knew their status before and after the CHiP visit, we assumed, first, that all those who knew they were HIV-positive self-reported this to the CHiP and, second, that among those whose HIV status was not known to the CHiP (because they declined to test with the CHiP and did not self-report that they were HIV-positive or that they had tested HIV-negative within the previous 3 mo), the HIV prevalence was the same as among those who accepted a test (Assumption A). Coverage against the first 90 target was estimated before and immediately after the CHiP visit, using data on HIV test uptake and results. Coverage against the second 90 target, based on those self-reporting that they were in HIV care or currently on ART among those known HIV-positive, was estimated before and immediately after the CHiP visit, and also at the end of the first annual round to take account of participants newly initiated on ART after the CHiP visit.

To estimate overall coverage against the first two 90 targets in the entire adult population of these communities, some additional assumptions were needed. This was because no data were available for households that did not consent to enumeration of household members and because, among households that consented to enumeration, the only information available for adults who did not consent to participate was their age and sex. We assumed, first, that HIV prevalence was the same in those consenting and not consenting to the intervention (Assumption B); second, that the proportion of HIV-positive individuals who knew their HIV status

prior to the CHiP intervention was the same in those consenting and not consenting (Assumption C); and, third, that the proportions of known HIV-positive individuals in care or on ART prior to the CHiP intervention were the same in those consenting and not consenting (Assumption D). All of the above assumptions (A to D) were assumed to apply within strata defined by sex, community, and age group.

We conducted two further analyses to explore the sensitivity of the above coverage estimates to the assumptions made. First, for each assumption (A to D), we assumed that the true proportion was equal to the assumed value or was increased or decreased by a factor of 1.25. We examined the range of coverage estimates resulting from these 81 (3⁴) combinations of values, and provide the minimum and maximum values together with the central estimate. Second, we made alternative assumptions about adults whose HIV status was not known to CHiPs: these alternative assumptions were that HIV prevalence was twice as high as among those who tested with the CHiPs, that the percentage who knew their HIV-positive status was the same as in those whose HIV status was known to CHiPs, and that ART uptake among those who knew their HIV-positive status was either the same or half the value as among those who self-reported as HIV-positive to the CHiPs.

Laboratory methods

CHiP participants were initially screened for HIV infection by fingerprick using the Alere Determine HIV-1/2 test (Alere International). Participants with a non-reactive test had no further testing. Participants with reactive results were further tested with the Uni-Gold HIV test (Trinity Biotech). If results were concordant, the participant was referred for care. If results were discordant, the testing was repeated. If the results were still discordant, the CHiPs would return 2 wk later for follow-up testing. If results were discordant again, a venous blood sample was collected and sent to the laboratory for further testing using the rapid tests and, if indicated, testing using the Abbott Architect HIV Ag/Ab Combo Assay. This test algorithm is in keeping with Zambian national guidelines.

Ethical considerations

During CHiP visits to households, all adults were asked for verbal informed consent to take part in the PopART intervention, which included the collection of data on the CHiP EDC. Those agreeing to a rapid HIV test gave written consent according to standard health care procedures for HIV testing and counselling in Zambia. The study and all the above procedures were approved by the ethics committees of the London School of Hygiene & Tropical Medicine and the University of Zambia.

Results

Cascade of care

A total of 48,583 households were visited during the first annual round in the four study communities in Zambia, close to 100% of all households listed during the community census prior to the study. Enumeration of individual household members was completed in 46,714 (96%) of the households visited, with a total of 121,130 adults aged 18 y and over (59,283 men and 61,847 women) (Fig 2). Of those enumerated, 77% of men and 90% of women consented to the PopART intervention. Of those consenting, 80% of men and 85% of women knew their HIV test status following the CHiP visit, because they self-reported that they were HIV-positive, because they accepted the offer of an HIV rapid test by the CHiP, or because they selfreported that they had tested HIV-negative in the previous 3 mo.



Fig 2. Cascade of care from enumeration of household members through ART initiation during the first annual round of the PopART intervention. ART, antiretroviral therapy.

https://doi.org/10.1371/journal.pmed.1002292.g002

Following the CHiP visit, a total of 4,138 men and 8,702 women were known to be HIV-positive, representing 11% and 18% of those who knew their HIV status among men and women, respectively. Of these, 2,133 men and 4,765 women reported that they were already in HIV care, and 1,941 and 4,308, respectively, were on ART. Among those not on ART, CHiPs referred a total of 2,053 men and 4,144 women to the local government health facility, including 5,783/ 5,847 (99%) of those reporting that they had never previously registered for HIV care.

Among the total of 6,197 adults who were not on ART at the time of the first household visit and were referred to HIV care, 42% (95% CI: 40%–43%) had initiated ART within 6 mo and 53% (95% CI: 52%–55%) within 12 mo, with very similar rates in men and women. Fig 3 shows survival plots for time from referral to linkage to care and ART initiation and indicates that by 3 mo only 41% (95% CI: 40%–42%) had linked to care and 32% (95% CI: 31%–33%) had initiated ART, but with additional individuals linking to care and starting ART each month up to at least 1 y.

Change in knowledge of HIV status and linkage to care following intervention

At the time of the CHiP visit, 18% of men and 26% of women who consented to the intervention self-reported that they were HIV-positive or that they had tested HIV-negative in the



Fig 3. Time from referral to linkage to care and ART initiation during the first annual round of the PopART intervention. Survival curves showing cumulative proportions linking to care or initiating ART following referral by community HIV-care providers. ART, antiretroviral therapy.

https://doi.org/10.1371/journal.pmed.1002292.g003

previous 3 mo and thus knew their current HIV test status (Table 1). After the CHiP visit, as a result of the HIV tests provided by the CHiPs, the proportion who knew their current test status increased to 80% and 85%, respectively.

Table 1. Change in knowledge of HIV status, linkage to care	, and ART uptake following the intervention among men and women.

Indicator	Before or after Round 1 CHiP visit	Men		Women	
		n/N	Percent	n/N	Percent
Knows HIV status/all who consented	Before ¹	8,305/45,399	18%	14,558/55,703	26%
	After ²	36,398/45,399	80%	47,089/55,703	85%
Known HIV-positive/all who consented	Before ¹	2,423/45,399	5%	5,309/55,703	10%
	After ²	4,138/45,399	9%	8,702/55,703	16%
In HIV care/all known HIV-positive	Before ²	2,133/4,138	52%	4,765/8,702	55%
	After ³	2,506/3,222	78%	5,528/7,043	78%
On ART/all known HIV-positive	Before ²	1,941/4,138	47%	4,308/8,702	49%
	After ³	2,309/3,222	72%	5,106/7,043	72%

¹Prior to annual round visit.

²Immediately following annual round visit.

³At end of Round 1 (restricted to adults still resident in the same zone of the community at end of Round 1 according to last recorded information from Round 1).

ART, antiretroviral therapy; CHiP, community HIV-care provider.

https://doi.org/10.1371/journal.pmed.1002292.t001

Table 2. Estimated proportions of HIV+ individuals who knew their status and who were on ART before the CHiP visit, immediately after the CHiP	
visit, and at the end of Round 1, in those consenting to the intervention.	

Indicator	Before CHiP visit	Immediately after CHiP visit	End Round 1 ¹	
Men				
Number HIV+ among those who consented to participate in Round 1 (estimated)	4,662	4,662	3,630	
Number (percent) who know HIV+ status	2,423 (52%)	4,138 (89%)	3,222 (89%)	
Number (percent) on ART	1,941 (80%)	1,941 (47%)	2,309 (72%)	
Overall proportion on ART	42%	42%	64%	
Women				
Number HIV+ among those who consented to participate in Round 1 (estimated)	9,499	9,499	7,688	
Number (percent) who know HIV+ status	5,309 (56%)	8,702 (92%)	7,043 (92%)	
Number (percent) on ART	4,308 (81%)	4,308 (49%)	5,106 (72%)	
Overall proportion on ART	45%	45%	66%	

¹End Round 1 numbers and percentages are restricted to adults still resident in the same zone of the community at the end of Round 1, according to the most recently available information.

ART, antiretroviral therapy; CHiP, community HIV-care provider.

https://doi.org/10.1371/journal.pmed.1002292.t002

Of the 4,138 men and 8,702 women known to be HIV-positive after the CHiP visit, 52% of men and 55% of women were already in HIV care, while 47% of men and 49% of women were already on ART. At the end of the first annual round, 3,222/4,138 (78%) of the known HIV-positive men and 7,043/8,702 (81%) of the known HIV-positive women were still resident in the same zone of the community according to the last available information. By the end of the first annual round of the intervention, the proportions of both men and women in care and on ART had increased to 78% and 72%, respectively.

Reaching the 90-90-90 targets

Data on knowledge of HIV-positive status and linkage to care were used to estimate coverage against the first two of the 90-90-90 targets at three time points, before and immediately after the CHiP visit and at the end of the first annual round of the intervention (Tables 2 and 3). These estimates were obtained based on the assumptions described in the Methods.

Table 3. Estimated proportions of HIV+ individuals who knew their status and who were on ART before the CHiP visit, immediately after the CHiP
visit, and at the end of Round 1, in total adult population.

Indicator	Before CHiP visit	Immediately after CHiP visit	End Round 1 ¹
Men			
Number HIV+ among total adult population in Round 1 (estimated)	6,649	6,649	5,206
Number (percent) who know HIV+ status	3,451 (52%)	5,166 (78%)	4,045 (78%)
Number (percent) on ART	2,769 (80%)	2,769 (54%)	2,975 (74%)
Overall proportion on ART	42%	42%	57%
Women			
Number HIV+ among total adult population in Round 1 (estimated)	11,037	11,037	8,958
Number (percent) who know HIV+ status	6,197 (56%)	9,590 (87%)	7,784 (87%)
Number (percent) on ART	5,036 (81%)	5,036 (53%)	5,715 (73%)
Overall proportion on ART	46%	46%	64%

¹End Round 1 numbers and percentages are restricted to adults still resident in the same zone of the community at the end of Round 1, according to the most recently available information.

ART, antiretroviral therapy; CHiP, community HIV-care provider.

https://doi.org/10.1371/journal.pmed.1002292.t003

Women





Fig 4. Estimated coverage compared with the first two 90 targets in those consenting to the PopART intervention. Dark blue bars show the estimated proportion of HIV+ adults who knew their status (first 90 target) and the estimated proportion of those who knew their HIV+ status who were on ART (second 90 target), pre-intervention. Red bars show the same estimated proportions, post-intervention. ART, antiretroviral therapy.

https://doi.org/10.1371/journal.pmed.1002292.g004

Among those consenting to the intervention, we estimate that prior to the intervention 52% of all HIV-positive men and 56% of all HIV-positive women knew their HIV-positive status. After the first annual round, these proportions increased to 89% of men and 92% of women, close to or exceeding the first 90 target (Fig 4).

Among those known HIV-positive prior to the CHiP visit, 80% of men and 81% of women reported that they were currently on ART. Immediately after the CHiP visit, this proportion decreased to 47% and 49% as a result of the increase in the denominator due to the large number of HIV-positive individuals newly diagnosed by the CHiPs. By the end of the annual round, however, the proportion on ART had increased to 72% and 72% in men and women, respectively.

Estimated coverage against the first two 90 targets in the entire adult population was somewhat lower because not everyone was enumerated or consented to the intervention. Based on the stated assumptions, we estimate that the proportion of HIV-positive individuals who knew their HIV-positive status increased from 52% of men and 56% of women before the intervention to 78% and 87%, respectively, after the first annual round. The proportion of known HIV-





Fig 5. Estimated coverage compared with the first two 90 targets extrapolated to the total adult population. The red line shows the 90% target for first two of the 90-90-90 targets. Dark blue bars show the estimated proportion of HIV+ adults who knew their status (first 90 target) and the estimated proportion of those who knew their HIV+ status who were on ART (second 90 target), pre-intervention. Red bars show the same estimated proportions, post-intervention. ART, antiretroviral therapy.

https://doi.org/10.1371/journal.pmed.1002292.g005

positive individuals on ART increased from 54% of men and 53% of women immediately after the CHiP visit to 74% and 73%, respectively, by the end of the annual round (Fig 5).

Table 2 also shows that the estimated overall proportion of HIV-positive adults on ART increased from 42% to 57% in men and from 46% to 64% in women, compared with the target of 81% based on the product of the first two of the 90-90-90 targets. S1 Fig shows an alternative display of coverage against the targets, similar to that used to report the Agence Nationale de Recherche sur le Sida et les Hépatites Virales 12249 Treatment as Prevention trial (ANRS 12249 TasP trial) [18].

Fig 6 shows the coverage estimates for the first and second 90 targets broken down by age group as well as sex. Prior to the CHiP intervention, knowledge of HIV-positive status and ART coverage were considerably lower in young men and women than in older age groups. HIV test uptake by young adults who participated in the intervention was high, so that by the end of the first annual round, knowledge of HIV-positive status was close to 90% in all age groups. In the entire population, estimated knowledge of HIV-positive status remained high among women,



Fig 6. First two 90 target estimates by sex and age group in those consenting to the PopART intervention and extrapolated to the total adult population. The red line shows the 90% target for first two of the 90-90-90 targets. The first 90 target is proportion of individuals with knowledge of HIV-positive status, and the second 90 target is proportion of known HIV-positive individuals on antiretroviral therapy. CHiP, community HIV-care provider.

https://doi.org/10.1371/journal.pmed.1002292.g006

but fell to around 70%–80% among young men because of the relatively large number who could not be contacted by CHiPs during household visits (17%, 20%, 24%, and 26% of men aged 18–19, 20–24, 25–29, and 30–34 y, respectively, were not contacted by CHiPs during Round 1). In both men and women, by the end of the first annual round, ART coverage among known HIV-positive individuals was relatively high in older age groups, at approximately 80% among men \geq 40 y and women \geq 35 y, but remained lower (around 50%–60%) in the youngest men and women. Fig 7 shows overall estimates of the proportion of HIV-positive adults on ART by age group and sex. After the first annual round, these estimates were approaching the 81% target in older adults, but were considerably lower in young men and women.

Sensitivity analysis

<u>S5 Table</u> shows two sensitivity analyses for the estimates of coverage for the first and second 90 targets. Focusing on coverage after the first annual round of the intervention, estimates were



Fig 7. Overall estimates of proportion of HIV+ adults on ART by age group and sex in those consenting to the PopART intervention and extrapolated to the total adult population. The red line shows the cumulative 81% target based on first two of the 90-90-90 targets. ART, antiretroviral therapy; CHiP, community HIV-care provider.

https://doi.org/10.1371/journal.pmed.1002292.g007

not sensitive to assumptions about adults who consented to the intervention. The sensitivity analyses produced a wider range of estimates when extrapolating to the entire adult population, although qualitative conclusions were broadly similar, with higher knowledge of HIV-positive status in women (82%–91%) than men (70%–84%) and with ART coverage among known HIV-positive individuals between 69%–79% in both men and women.

Discussion

This study showed substantial increases in knowledge of HIV status and ART coverage after the first annual round of a UTT intervention delivered through a home-based service linked to government health facilities providing care and treatment for HIV. Among those consenting to the intervention, knowledge of current HIV status increased from around 20% to greater than 80%, while ART coverage among known HIV-positive individuals increased from around 50% to 70%. In the entire adult population, the estimated proportion of HIV-positive individuals who knew their status (first 90 target) increased from 52% to 78% in men and from 56% to 87% in women, while the proportion of known HIV-positive individuals on ART (second 90 target) increased from 54% after the CHiP visit to 74% by the end of the round in men and from 53% to 73% in women. The estimated overall proportion of HIV-positive adults on ART, irrespective of whether they knew their HIV status, increased from 44% to 61%, compared with the 81% UNAIDS target for the first two 90 targets.

The concept of treatment as prevention has revitalised efforts to bring the HIV epidemic under effective control. The rationale for the concept is strong, and if high coverage can be achieved, the evidence suggests that the incidence of new infections should be reduced substantially. As a result, it now seems realistic to consider scenarios in which HIV prevalence and incidence are reduced in the long term to low levels, such that HIV/AIDS no longer constitutes a major public health problem.

These potential benefits will not be achieved, however, unless high rates of coverage can be achieved and sustained. There is increasing recognition that this will be challenging, both because of the many barriers to testing and care in settings with severe resource constraints and because high uptake requires the long series of steps in the cascade of care to be successfully negotiated [19–22]. The 90-90-90 targets, if they can be achieved, imply that only 73% of individuals living with HIV will be virally suppressed and therefore unlikely to transmit the virus to others [15]. If coverage is reduced to 80% at each step, the proportion virally suppressed falls to 51%. The problem is compounded because each of the 90-90-90 targets itself involves multiple steps. For example, the first 90 target requires that an individual is offered a test (at regular intervals), accepts the test, and receives the correct result. The second 90 target requires that the individual is effectively linked to care, and then attends repeated clinic visits until ART is prescribed and commenced. The third 90 target requires retention in care, high adherence to ART, and a regimen that is effective in achieving viral suppression in that individual. Seen in this light, the 90-90-90 targets will not be easy to achieve.

There have been relatively few attempts to estimate coverage against the targets on a large scale in resource-poor settings in sub-Saharan Africa [23–28]. We have used data from the largest ongoing programme to deliver UTT at the population level to provide such estimates. We find that after one annual round of the PopART intervention in four large communities in Zambia, we have come close to achieving the first 90 target in women and are approaching it in men. The second 90 target has proven more challenging, and although we have increased the estimated proportion of known HIV-infected individuals on ART from around 53% to 73%, the coverage needs to be increased. Data on the third 90 target, relating to retention on ART and viral suppression, will be obtained during the second annual round of the intervention.

Our findings agree with other studies in showing that a house-to-house service is able to achieve high rates of testing and knowledge of HIV status among community members [29]. Detailed results on test uptake will be reported elsewhere, but indicate that coverage of HIV testing is systematically higher among women than men. While men are only slightly more likely than women to decline a test when it is offered by the CHiPs, a significant challenge is that they are often not found at home during house-to-house visits. Reductions in the high incidence of HIV among women, particularly young women, will be limited unless their male partners are effectively tested, linked to care, and virally suppressed on ART [30]. Other studies have also found poorer coverage among men with home-based testing, and have suggested that additional testing strategies such as mobile or occupational testing facilities or self-testing may be needed to achieve high coverage in this group [31].

As in other programmes, the time taken for those diagnosed HIV-positive to link to care and commence ART has been longer than anticipated [19,32,33]. Our initial target was for ART to be commenced in 80% of individuals within 3 mo of referral. In practice, we found that slightly over 50% of both men and women commenced ART within 12 mo of referral.

While the survival curves show that more and more individuals start ART over time, we need to speed up the rates of linkage and commencement of ART, and this was identified as a major focus of the CHiP intervention for the second annual round. Impact on HIV incidence will be limited unless HIV-positive individuals can be linked to care and initiated on ART rapidly, with lifelong retention and adherence to ART.

Of key importance were our findings on differential coverage across age groups. After the first annual round of intervention, knowledge of HIV status was high across all age groups in women and in older men, but was considerably lower among young men. Test uptake was high among young men who consented to the intervention, but a relatively large number of young men could not be contacted by CHiPs. ART coverage at the end of the first annual round among known HIV-positive individuals was considerably lower in young men and women than in older men and women, consistent with previous research [34]. Linkage to care and initiation of ART among those referred by CHiPs were slightly slower among younger adults, and far fewer young adults were already on ART at the start of the intervention, and this depressed coverage for the second 90 target. These findings emphasise the importance of contacting and engaging young men and women, and linking them to care as rapidly as possible. Full data on uptake by age and sex will be published separately.

Three other community-randomised trials in sub-Saharan Africa are measuring the impact of UTT interventions on HIV incidence: the ANRS 12249 TasP trial in South Africa [18], the Sustainable East Africa Research on Community Health (SEARCH) trial in Kenya and Uganda [35], and the Botswana Combination Prevention Project (BCPP) trial in Botswana [26]. Results from the first of these, the ANRS 12249 TasP trial, have recently been reported [36]. In this study in KwaZulu-Natal, South Africa, six-monthly home-based testing resulted in high uptake of HIV testing among those contacted, and high rates of viral suppression 12 mo after ART initiation. As in our study, linkage to care and ART initiation (at clinics provided by the study) were slower than expected, with only 36% accessing care within 6 mo of referral. Analysis of HIV incidence showed no evidence of impact. Importantly, however, the home-based programme for HIV testing and linkage to care was delivered to both intervention and control arms of the study, which differed only in the threshold for ART initiation. Overall coverage of ART was similar in the two study arms, and so it is unsurprising that no impact on HIV incidence was observed. In contrast, the HPTN 071 (PopART) trial will compare HIV incidence in intervention communities receiving the full UTT intervention with that in control communities receiving current standard of care (not including home-based testing).

Our study has a number of limitations. Data on coverage are only collected on those who participate in the CHiP intervention, and in order to obtain estimates for the entire adult population, we had to make assumptions about those not participating. We also rely on self-report of known HIV-positive status and whether individuals are in care or on ART, although documentary evidence is sought where possible. Another limitation is that our estimates of coverage ignore any new HIV seroconversions occurring between the initial annual round visit and the end of the round, although these are likely to have only a small effect, given the average duration of follow-up and expected HIV incidence. A further limitation is that data were collected only on adults aged 18 y and over. Although these will encompass a large proportion of those with HIV infection, we know that there is a continuing high incidence and prevalence in young people in many settings. Following ethical review, we are now able to collect data on minors, subject to parental consent, and so in future will be able to document uptake in this important age group.

Strengths of our study are that we have collected data systematically for most households and a large proportion of adults in these large communities, and have stated our methods and assumptions clearly. We have also carried out analyses to examine the sensitivity of our estimates to the assumptions made. We believe that comparisons of coverage estimates between countries, settings, and programmes are susceptible to significant bias and distortion due to different availability of data and assumptions made in the analyses, and some harmonisation of such analyses is called for. The Universal Test and Treat Trials Consortium (UT³C), which comprises investigators from five randomised trials of UTT interventions in sub-Saharan Africa, is planning joint cross-study analyses to further explore this issue.

This paper reports uptake following the first year of the PopART intervention. A strength of this intervention is that household visits are made to all households during each annual round, as well as repeat visits during each year as appropriate. This approach offers the opportunity to continuously increase coverage, by accessing individuals who were previously not at home, offering tests to those who initially declined testing, and making special efforts to encourage linkage to care where this has not already occurred. After the first year, estimated overall coverage against the first two 90 targets combined has increased from 44% to 61%, indicating the gap still to be filled to reach the target of 81%. It will be important to document how uptake evolves over time, and we plan to report this after the second annual round of the intervention. The initial data, however, demonstrate that a community-wide intervention provided through a house-to-house service in communities with severe generalised HIV epidemics may be effective in achieving or approaching the 90-90-90 targets.

Supporting information

S1 STROBE Checklist. (DOC)

S1 Data. Aggregate dataset. (XLSX)

S2 Data. Definition of variables in aggregate dataset. (DOCX)

S1 Fig. Estimated uptake in total adult population—alternative display. (PPTX)

S1 Table. Estimate of the proportion of HIV+ women who knew their HIV+ status, among those aged 25–29 y who consented to participate in the CHiP intervention. (DOCX)

S2 Table. Estimate of uptake of ART, among HIV+ women aged 25–29 y who consented to participate in the CHiP intervention and who knew their HIV+ status following the Round 1 annual household visit.

(DOCX)

S3 Table. Estimate of the proportion with knowledge of HIV+ status among all HIV+ women in the population aged 25–29 y (extrapolation to women who did not participate in the CHiP intervention in Round 1). (DOCX)

S4 Table. Estimate of uptake of ART among all HIV+ women in the population aged 25–29 y who knew their HIV+ status (extrapolation to women who did not participate in the CHIP intervention in Round 1).

(DOCX)

S5 Table. Summary table of estimates for the first two 90-90-90 targets, Zambia Round 1; adults aged \geq 18 y at time of annual round visit; sensitivity analysis. (DOCX)

Acknowledgments

We are grateful to all members of the HPTN 071 (PopART) Study Team, and to the study participants and their communities, for their contributions to the research.

The HPTN 071 (PopART) Study Team: James Hargreaves (London School of Hygiene & Tropical Medicine, London, UK), Deborah Watson-Jones (London School of Hygiene & Tropical Medicine, London, UK), Peter Godfrey-Faussett (London School of Hygiene & Tropical Medicine, London, UK), Katharina Hauck (Imperial College London, London, UK), Peter C. Smith (Imperial College London, London, UK), Anne Cori (Imperial College London, London, UK), Mike Pickles (Imperial College London, London, UK), Nomtha Mandla (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Blia Yang (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Anelet James (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Redwaan Vermaak (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Nozizwe Makola (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Graeme Hoddinott (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Vikesh Naidoo (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Virginia Bond (London School of Hygiene & Tropical Medicine, London, United Kingdom, and Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Musonda Simwinga (Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Alwyn Mwinga (Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Barry Kosloff (Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Mohammed Limbada (Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Justin Bwalya (Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Chepela Ngulube (Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Christophe Fraser (Nuffield Department of Medicine, Oxford University, Oxford, UK), Susan Eshleman (Department of Pathology, Johns Hopkins University School of Medicine, Baltimore, Maryland, US), Yaw Agyei (Department of Pathology, Johns Hopkins University School of Medicine, Baltimore, Maryland, United States), Vanessa Cummings (Department of Pathology, Johns Hopkins University School of Medicine, Baltimore, Maryland, US), Denni Catalano (Department of Pathology, Johns Hopkins University School of Medicine, Baltimore, Maryland, US), Lynda Emel (Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, Washington, US), Lisa Bunts (Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, Washington, US), Heather Noble (Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, Washington, US), David Burns (Division of AIDS, National Institute of Allergy and Infectious Diseases, National Institutes of Health, Bethesda, Maryland, US), Alain Kouda (Division of AIDS, National Institute of Allergy and Infectious Diseases, National Institutes of Health, Bethesda, Maryland, US), Niru Sista (FHI 360, Durham, North Carolina, US), Ayana Moore (FHI 360, Durham, North Carolina, US), Rhonda White (FHI 360, Durham, North Carolina, US), Tanette Headen (FHI 360, Durham, North Carolina, US), Eric Miller (FHI 360, Durham, North Carolina, US), Kathy Hinson (FHI 360, Durham, North Carolina, US), Sten Vermund (Yale University, New Haven, Connecticut, US), Mark Barnes (Ropes & Gray, Boston, Massachusetts, US), Lyn Horn (Desmond Tutu TB Centre, Stellenbosch University, Stellenbosch, South Africa), Albert Mwango (Zambart, University of Zambia School of Medicine, Lusaka, Zambia), Megan Baldwin (Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, Washington, US), Shauna Wolf (Department of Pathology, Johns Hopkins University School of Medicine, Baltimore, Maryland, US), and Erin Hughes (Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, Washington, US).

The content herein is solely the responsibility of the authors and does not necessarily represent the official views of the National Institute of Allergy and Infectious Diseases, National Institute of Mental Health, National Institute on Drug Abuse, the US President's Emergency Plan for AIDS Relief, the International Initiative for Impact Evaluation, or the Bill & Melinda Gates Foundation.

Author Contributions

Conceptualization: RH SFl KSa NB HA SFi. **Data curation:** SFl AS.

Formal analysis: RH SFl AS.

Funding acquisition: RH KSa NB HA SFi.

Investigation: KSh PB SG NB HA.

Methodology: RH SFi AS KSa DD.

Project administration: KSh PB SG HA.

Software: AS.

Supervision: RH WE-S NB HA SFi.

Validation: DD EP-M WE-S.

Writing - original draft: RH.

Writing - review & editing: RH SFl AS KSh PB KSa SG DD EP-M WE-S NB HA SFi.

References

- 1. Joint United Nations Programme on HIV/AIDS. The gap report: beginning of the end of the AIDS epidemic. Geneva: Joint United Nations Programme on HIV/AIDS; 2014 [cited 2016 Jan 17]. Available from: http://www.unaids.org/sites/default/files/media_asset/UNAIDS_Gap_report_en.pdf.
- Attia S, Egger M, Muller M, Zwahlen M, Low N. Sexual transmission of HIV according to viral load and antiretroviral therapy: systematic review and meta-analysis. AIDS. 2009; 23(11):1397–404. https://doi. org/10.1097/QAD.0b013e32832b7dca PMID: 19381076
- Cohen MS, Chen YQ, McCauley M, Gamble T, Hosseinipour MC, Kumarasamy N, et al. Prevention of HIV-1 infection with early antiretroviral therapy. N Engl J Med. 2011; 365(6):493–505. <u>https://doi.org/10.1056/NEJMoa1105243</u> PMID: 21767103
- Wawer MJ, Gray RH, Sewankambo NK, Serwadda D, Li X, Laeyendecker O, et al. Rates of HIV-1 transmission per coital act, by stage of HIV-1 infection, in Rakai, Uganda. J Infect Dis. 2005; 191(9):1403–9. https://doi.org/10.1086/429411 PMID: 15809897
- Quinn TC, Wawer MJ, Sewankambo N, Serwadda D, Li C, Wabwire-Mangen F, et al. Viral load and heterosexual transmission of human immunodeficiency virus type 1. Rakai Project Study Group. N Engl J Med. 2000; 342(13):921–9. https://doi.org/10.1056/NEJM200003303421303 PMID: 10738050
- Cori A, Ayles H, Beyers N, Schaap A, Floyd S, Sabapathy K, et al. HPTN 071 (PopART): a cluster-randomized trial of the population impact of an HIV combination prevention intervention including universal testing and treatment: mathematical model. PLoS ONE. 2014; 9(1):e84511. https://doi.org/10.1371/ journal.pone.0084511 PMID: 24454728
- Dodd PJ, Garnett GP, Hallett TB. Examining the promise of HIV elimination by 'test and treat' in hyperendemic settings. AIDS. 2010; 24(5):729–35. https://doi.org/10.1097/QAD.0b013e32833433fe PMID: 20154580
- 8. Granich RM, Gilks CF, Dye C, De Cock KM, Williams BG. Universal voluntary HIV testing with immediate antiretroviral therapy as a strategy for elimination of HIV transmission: a mathematical model. Lancet. 2009; 373(9657):48–57. https://doi.org/10.1016/S0140-6736(08)61697-9 PMID: 19038438

- Lima VD, Johnston K, Hogg RS, Levy AR, Harrigan PR, Anema A, et al. Expanded access to highly active antiretroviral therapy: a potentially powerful strategy to curb the growth of the HIV epidemic. J Infect Dis. 2008; 198(1):59–67. https://doi.org/10.1086/588673 PMID: 18498241
- Granich R, Gupta S, Hersh B, Williams B, Montaner J, Young B, et al. Trends in AIDS deaths, new infections and ART coverage in the top 30 countries with the highest AIDS mortality burden; 1990–2013. PLoS ONE. 2015; 10(7):e0131353. https://doi.org/10.1371/journal.pone.0131353 PMID: 26147987
- Montaner JS, Lima VD, Barrios R, Yip B, Wood E, Kerr T, et al. Association of highly active antiretroviral therapy coverage, population viral load, and yearly new HIV diagnoses in British Columbia, Canada: a population-based study. Lancet. 2010; 376(9740):532–9. https://doi.org/10.1016/S0140-6736(10) 60936-1 PMID: 20638713
- Montaner JS, Lima VD, Harrigan PR, Lourenco L, Yip B, Nosyk B, et al. Expansion of HAART coverage is associated with sustained decreases in HIV/AIDS morbidity, mortality and HIV transmission: the "HIV treatment as prevention" experience in a Canadian setting. PLoS ONE. 2014; 9(2):e87872. https://doi. org/10.1371/journal.pone.0087872 PMID: 24533061
- Tanser F, Barnighausen T, Grapsa E, Zaidi J, Newell ML. High coverage of ART associated with decline in risk of HIV acquisition in rural KwaZulu-Natal, South Africa. Science. 2013; 339(6122):966–71. https://doi.org/10.1126/science.1228160 PMID: 23430656
- Joint United Nations Programme on HIV/AIDS. Treatment 2015. Geneva: Joint United Nations Programme on HIV/AIDS; 2012 [cited 2016 Jan 11]. Available from: http://www.unaids.org/sites/default/ files/media_asset/JC2484_treatment-2015_en_1.pdf.
- 15. Joint United Nations Programme on HIV/AIDS. 90-90-90: an ambitious treatment target to help end the AIDS epidemic. Geneva: Joint United Nations Programme on HIV/AIDS; 2014 [cited 2016 Jan 11]. Available from: http://www.unaids.org/sites/default/files/media_asset/90-90-90_en_0.pdf.
- 16. Joint United Nations Programme on HIV/AIDS. Fast-track: ending the AIDS epidemic by 2030. Geneva: Joint United Nations Programme on HIV/AIDS; 2014 [cited 2016 Jan 11]. Available from: http://www.unaids.org/sites/default/files/media_asset/JC2686_WAD2014report_en.pdf.
- Hayes R, Ayles H, Beyers N, Sabapathy K, Floyd S, Shanaube K, et al. HPTN 071 (PopART): rationale and design of a cluster-randomised trial of the population impact of an HIV combination prevention intervention including universal testing and treatment—a study protocol for a cluster randomised trial. Trials. 2014; 15:57. https://doi.org/10.1186/1745-6215-15-57 PMID: 24524229
- Iwuji CC, Orne-Gliemann J, Tanser F, Boyer S, Lessells RJ, Lert F, et al. Evaluation of the impact of immediate versus WHO recommendations-guided antiretroviral therapy initiation on HIV incidence: the ANRS 12249 TasP (Treatment as Prevention) trial in Hlabisa sub-district, KwaZulu-Natal, South Africa: study protocol for a cluster randomised controlled trial. Trials. 2013; 14:230. https://doi.org/10.1186/ 1745-6215-14-230 PMID: 23880306
- Barnabas R, van Rooyen H, Tumwisigye E, Brantley J, Baeten J, van Heerden A, et al. Uptake of antiretroviral therapy and male circumcision after community-based HIV testing and strategies for linkage to care versus standard clinic referral: a multisite, open-label, randomised controlled trial in South Africa and Uganda. Lancet HIV. 2016; 3(5):e212–20. https://doi.org/10.1016/S2352-3018(16)00020-5 PMID: 27126488
- McNairy ML, El-Sadr WM. The HIV care continuum: no partial credit given. AIDS. 2012; 26(14):1735–8. https://doi.org/10.1097/QAD.0b013e328355d67b PMID: 22614888
- Rosen S, Fox MP. Retention in HIV care between testing and treatment in sub-Saharan Africa: a systematic review. PLoS Med. 2011; 8(7):e1001056. https://doi.org/10.1371/journal.pmed.1001056 PMID: 21811403
- Sabapathy K, Hayes R. Linkage to care: a step on the path, but not the destination. Lancet HIV. 2016; 3 (5):e193–4. https://doi.org/10.1016/S2352-3018(16)00038-2 PMID: 27126481
- Maman D, Chilima B, Masiku C, Ayouba A, Masson S, Szumilin E, et al. Closer to 90-90-90. The cascade of care after 10 years of ART scale-up in rural Malawi: a population study. J Int AIDS Soc. 2016; 19(1):20673. https://doi.org/10.7448/IAS.19.1.20673 PMID: 26894388
- Levira F, Agnarson AM, Masanja H, Zaba B, Ekstrom AM, Thorson A. Antiretroviral treatment coverage in a rural district in Tanzania—a modeling study using empirical data. BMC Public Health. 2015; 15:195. https://doi.org/10.1186/s12889-015-1460-8 PMID: 25884639
- 25. Levi J, Raymond A, Pozniak A, Vernazza P, Kohler P, Hill A. Can the UNAIDS 90-90-90 target be achieved? A systematic analysis of national HIV treatment cascades. BMJ Global Health. 2016; 1: e000010.
- Gaolathe T, Wirth KE, Holme MP, Makhema J, Moyo S, Chakalisa U, et al. Botswana's progress toward achieving the 2020 UNAIDS 90-90-90 antiretroviral therapy and virological suppression goals: a population-based survey. Lancet HIV. 2016; 3(5):e221–30. <u>https://doi.org/10.1016/S2352-3018(16)00037-0</u> PMID: 27126489

- 27. Iwuji CC, Orne-Gliemann J, Larmarange J, Okesola N, Tanser F, Thiebaut R, et al. Uptake of homebased HIV testing, linkage to care, and community attitudes about ART in Rural KwaZulu-Natal, South Africa: descriptive results from the first phase of the ANRS 12249 TasP cluster-randomised trial. PLoS Med. 2016; 13(8):e1002107. https://doi.org/10.1371/journal.pmed.1002107 PMID: 27504637
- 28. Petersen M, Balzer L, Kwarsiima D, Sang N, Chamie G, Ayieko J, et al. SEARCH test and treat study in Uganda and Kenya exceeds the UNAIDS 90-90-90 cascade target by achieving 81% population-level viral suppression after 2 years. Abstract WEAC0106LB2016. 21st International AIDS Conference; 18–22 Jul 2016; Durban, South Africa. Geneva: International AIDS Society; 2016 [cited 2017 Mar 28]. Available from: http://programme.aids2016.org/Abstract/Abstract/10514.
- 29. Sabapathy K, Van den Bergh R, Fidler S, Hayes R, Ford N. Uptake of home-based voluntary HIV testing in sub-Saharan Africa: a systematic review and meta-analysis. PLoS Med. 2012; 9(12):e1001351. https://doi.org/10.1371/journal.pmed.1001351 PMID: 23226107
- Dellar RC, Dlamini S, Karim QA. Adolescent girls and young women: key populations for HIV epidemic control. J Int AIDS Soc. 2015; 18(2 Suppl 1):19408.
- Sharma M, Ying R, Tarr G, Barnabas R. Systematic review and meta-analysis of community and facility-based HIV testing to address linkage to care gaps in sub-Saharan Africa. Nature. 2015; 528(7580): S77–85. https://doi.org/10.1038/nature16044 PMID: 26633769
- 32. Barnabas RV, van Rooyen H, Tumwesigye E, Murnane PM, Baeten JM, Humphries H, et al. Initiation of antiretroviral therapy and viral suppression after home HIV testing and counselling in KwaZulu-Natal, South Africa, and Mbarara district, Uganda: a prospective, observational intervention study. Lancet HIV. 2014; 1(2):e68–76. https://doi.org/10.1016/S2352-3018(14)70024-4 PMID: 25601912
- 33. Hoffmann C, Mabuto T, Ginindza S, Fielding K, Kubeka G, Dowdy D, et al. A randomized trial to accelerate HIV care and art initiation following HIV diagnosis. Abstract 113LB. 23rd Conference on Retroviruses and Opportunistic Infections; 22–25 Feb 2016; Boston, Massachusetts, US. San Francisco: CROI Foundation; 2016 [cited 2017 Mar 28]. Available from: http://www.croiconference.org/sessions/ randomized-trial-accelerate-hiv-care-and-art-initiation-following-hiv-diagnosis.
- 34. Feldacker C, Johnson D, Hosseinipour M, Phiri S, Tweya H. Who starts? Factors associated with starting antiretroviral therapy among eligible patients in two, public HIV clinics in Lilongwe, Malawi. PLoS ONE. 2012; 7(11):e50871. https://doi.org/10.1371/journal.pone.0050871 PMID: 23226413
- 35. Chamie G, Clark TD, Kabami J, Kadede K, Ssemmondo E, Steinfeld R, et al. A hybrid mobile approach for population-wide HIV testing in rural east Africa: an observational study. Lancet HIV. 2016; 3(3): e111–9. https://doi.org/10.1016/S2352-3018(15)00251-9 PMID: 26939734
- 36. Iwuji C, Orne-Gliemann J, Balestre E, Larmarange J, Thiebaut R, Tanser F, et al. The impact of universal test and treat on HIV incidence in a rural South African population: ANRS 12249 TasP trial, 2012–2016. Abstract FRAC0105LB2016. 21st International AIDS Conference; 18–22 Jul 2016; Durban, South Africa. Geneva: International AIDS Society; 2016 [cited 2017 Mar 28]. Available from: http://programme.aids2016.org/Abstract/Abstract/10537.